CSCE 970: Pattern Recognition Spring Semester, 2001

Homework Assignment 0

January 8, 2001 Due Date: January 22

1. (10 pts) Send e-mail to sscott@cse with exactly the following in the subject

CSCE 970 hwk 0

and exactly the following in the body of the message:

970 SSN addr alias name

where name is your name, SSN is your student ID number, addr is your e-mail address (you may leave off the .unl.edu), and alias is an alias that you will use to identify yourself when I post grades on the web page. (I use the alias in lieu of social security numbers.) If you do not wish to have your grades posted, make your alias “NO”. Also, do not place any whitespace within any of the fields.

Please send me this e-mail ASAP so you will be placed on the class e-mail list. 2. (10 pts) E-mail me a picture of yourself. This is to help me learn your names and maintain an on-line roster. If there are multiple people in the photo, clearly indicate which one is you.